%%  For format this file, run the following commands:
%%  
%%  latex agsmtst.tex
%%  bibtex agsmtst.tex
%%  latex agsmtst.tex
%%  latex agsmtst.tex
%%  makeindex -o agsmtst.and agsmtst.adx
%%  latex agsmtst.tex

\documentstyle[index,autind,harvard]{article}

\newindex{aut}{adx}{and}{Name Index}

\begin{document}

\adjustcitation\cite

\adjustcitation{\citeasnoun}

%{\tt\meaning\cite}

%{\tt\meaning\citeasnoun}

\nocite*

\cite{CantoneCP90,Adamowicz92}

\cite[page 21]{AdlemanMa76}

\citeasnoun{CaludeP83}

\citeasnoun[foo]{Baker68}

\newpage

\bibliographystyle{xagsm}
\bibliography{test}

\printindex[aut]

\end{document}
